Pbar Complex Summary (week

* AD
* Horn in fault.
* Local Reset. Impossible to do Reset remotely.
* Water leak on the C10 cryo-pump.
* Fixed Friday.
* Water leak on ECooler.
* Changed a water pipe.
e PSB + 2 turns

* 6% more protons on target, we get +5% at the exit of AD
e 1800e10 from PS, 1600e10 on target, 3.7e7 out AD
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* ELENA

» Several correctors OFF in Alpha line.
* First line has exchanged the control board.

* Control board of the filament of the source has been repaired.
* Tests ongoing

* Glitch on a "Gbar" vacuum pump triggered the closure of the fast vacuum
valve on LNE50

* Piquet VAC opened it

* New AD + ELENA vistar deployed.
 AD / ELENA vistar details : - PS Operations - Controls Wikis (cern.ch)



https://wikis.cern.ch/pages/viewpage.action?pageId=208569700
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|| Stat of the Week
fAnti-Pmtnns extracted | AD efficiencies | AD Target |

Anti-protons production - 2022 Week 44

AD: 1.51E11 in total over 4547 cycles, (DE.BCT7049)
ELEMNA: 1.52E11 in total over 4424 cycles, (LNRAPULE.O227)
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